Author’s response to reviews

Title: Workforce Patterns in the Prevention of Mother to Child Transmission of HIV in Côte d'Ivoire: A Qualitative Model

Authors:

Brianne Huffstetler Rowan (briannehuffstetler@gmail.com)

Julia Robinson (juliarob@uw.edu)

Samuel Granato (sgranato@uw.edu)

Claire Konan Bla (blaclairekonan@yahoo.fr)

Seydou Kouyaté (seydou.kouyate@healthallianceinternational.org)

Guy Djety 0028guydjety@yahoo.fr)

Kouamé Abo (kwagny@gmail.com)

Ahoua Koné (ahoua.kone@hai-ci.org)

Stephen Gloyd (gloyd@uw.edu)

Version: 2 Date: 19 Dec 2017

Author’s response to reviews:

Thank you for your careful review.

Reviewer #1: I have restructured the requested sentence into 2 sentences and fixed the grammar. Thank you for noting that error.

Reviewer #2: I have clarified with the country policy that in fact they followed the WHO Option A strictly and did not make country-specific changes regarding infant treatment (as my prior statement has suggested). I have updated the paragraph to reflect standard WHO policy regarding Option A (thus clarifying that all infants did in fact receive prophylaxis for at least 4-6 weeks, and that the differences were only in length of infant therapy and in whether the mothers were continuing ARV therapy).

Thank you,

Brianne H. Rowan